Author's response to reviews

Title: Liquid and vapour-phase antifungal activities of selected essential oils against Candida albicans: Microscopic observations and chemical characterization of Cymbopogon citratus

Authors:

Amit K Tyagi (amittyagiiitd@gmail.com)
Anushree Malik (anushree_malik@yahoo.com)

Version: 7 Date: 9 November 2010

Author's response to reviews: see over
Dear Editor,

Please find herewith the response to “Further Formatting Changes” suggested on the above-referred manuscript. The required changes have been made and the revised manuscript is being uploaded.

Kind regards
Amit Tyagi

--------------------------------------------------------------------------------------------------

**Formatting changes to be made:**

- **Major revisions (we require the author to make these changes)**

  **Author list:** the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name'(e.g. Joe F Bloggs). Currently they do not match.

  **Done**

- **Footnotes:** We are unable to accommodate footnotes in the full text version of the manuscript. We suggest that you incorporate the footnotes into the text of the manuscript. Alternatively, you may wish to make an appendix section containing your footnotes.

  **All footnotes have been incorporated into the text of the manuscripts.**

- **References:** References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Reference 42 is currently cited out of numerical order. Please ensure that all references are cited in consecutive numerical order in the manuscript.

  **Done**

Please make sure you use the latest version of the main file as we have made several formatting changes and using a previous unformatted version will delay publication.

- **Typography:** Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

  **Thank you**

--------------------------------------------------------------------------------------------------